

- Operations
 - Observations:

HAH/4/2/2008

• Pulsar & Long baseline (See later talks)







- Stations
 - Firmware update RSP boards CS001/016/010 * configuration (Load balancing)
 - ⊕ CS001T RCU 0, 1 repaired
 - Plan for distributing SW&configurations
 - Proposal Arno being reviewed by Ruud







- Network
 - Fiber with high data packet loss
 - Switched SFP unit
 - Lightpath LOFAR Amsterdam
 - To be done







- CEP
 - Distribute datasets over storage nodes
 - Waiting for cleaning up of some storage nodes!
 - Planned for next week
 - Work on distributing software images
 - New lioff systems (old input cluster)





